Author's response to reviews

Title: Herbal medicine use among urban residents in Lagos, Nigeria

Authors:

Ibrahim Adekunle Oreagba (oreagbai@yahoo.com)
Kazeem Adeola Oshikoya (med_modhospital@yahoo.com)
Mercy Amachree (feltina2002@yahoo.com)

Version: 6 Date: 22 November 2011

Author's response to reviews:

Typo errors have been identified and corrected. The section of the methods where the questionnaire was pretested and validated was repeated, the repeat sentence has been deleted.